Author’s response to reviews

Title: Synovial sarcoma of the floor of the mouth: A rare case report

Authors:
Yannan Wang (wangyannan@csu.edu.cn)
Feiya Zhu (physics4@csu.edu.cn)
Kai Wang (188102160@csu.edu.cn)

Version: 2 Date: 11 Nov 2019

Author’s response to reviews:

Dear reviewers and editors,

Thank you for your comments and we are very appreciated.

Here is the revised manuscript with some corrections. And, we would like to thank Branden Wang for constructive writing and editing support of the manuscript.

Respond to the reviewers comments:

Harsha Lal De Silva, BDS, MS, FDSRCS. FFDRCS(Reviewer 1):
Thank you for your suggestions.

Response:
R1C1: Does this mean you approached the submandibular area (Inferior to the mylohyoid) via an intra-oral excision? or conversely there is an error in your description and you are actually referring to sublingual (on th oral side of the mylohyoid)?

Authors response R1C1: We are sorry for the confusion. It means that an intra-oral excision in the floor of mouth of the right submandibular region (Inferior to the mylohyoid).

R1C2: What is the total length of follow-up so far? This is important before you claim success for the treatment provided. you show these scans?

Authors response R1C2: So far, the total length of follow-up is 46 months (page 3, line 42).
Jayantha Weerasinghe (Reviewer 2):
Thank you very much.

Thank you for all your suggestions and recommendations.

Best,

Kai Wang